\begin{table}%t1
\caption{Representative parameters for SV Vulpeculae. \label{tbl-1}}
%\centering
\par
\begin{tabular}{lll}
\hline
\hline
Parameter &Value &Source \\
\hline
Period &$45\fd0121$ &Khurkarkin et~al. (\cite{kh85}) \\
$\langle V \rangle$ &7.209 &Berdnikov (\cite{be02}) \\
$\langle B \rangle - \langle V \rangle$ &1.462 &Berdnikov (\cite{be02}) \\
Blue amplitude &1.63 &Berdnikov (\cite{be02}) \\
Reddening $E_{B-V}$ &$0.45 \pm 0.01$ &Turner (\cite{tu84}) \\
Progenitor mass &$\sim$$17~M_{\odot}$ &Turner (\cite{tu96}) \\
Mean radius &$201.0$ $\pm$ $6.0~R_{\odot}$ & Turner \& Burke (\cite{tb02}) \\
Mean $T_{\rm eff}$ &4830 K &Turner \& Burke (\cite{tb02}) \\
Luminosity &$1.98$ $\times$ $10^4~L_{\odot}$ &Turner \& Burke (\cite{tb02}) \\
\hline
\end{tabular}
\end{table}